Author’s response to reviews

Title: Sarcomatoid renal cell carcinoma: A case report and literature review

Authors:

Xiang Liang (liangx2003@126.com)
Yupin Liu (1034069077@qq.com)
Pengcheng Ran (cmurpc@163.com)
Meili Tang (merytang@126.com)
Changlei Xu (656704860@qq.com)
Yazhen Zhu (28905709@qq.com)

Version: 3 Date: 02 Nov 2017

Author’s response to reviews:

BNEP-D-16-00429R1

Sarcomatoid renal cell carcinoma: A case report and literature review

Xiang Liang; Yupin Liu; Pengcheng Ran; Meili Tang; Changlei Xu; Yazhen Zhu

BMC Nephrology

Dear BMC Nephrology Editorial Office,

Thanks you very much for your comments and suggestions.

We have revised the manuscript, according to the comments and suggestions of reviewers and editor, and responded, point by point to, the comments as listed below. All changes to the manuscript are indicated in the text by highlighting.

With the help of a professional writer, we have re-edited the paper.

I would like to re-submit this revised manuscript to BMC Nephrology, and hope it is acceptable for publication in the journal.

Looking forward to hearing from you soon.
With kindest regards,

Yours Sincerely

M.D. liu

Replies to Reviewers and Editor

First of all, we thank both reviewers and editor for their positive and constructive comments and suggestions.

Replies to Editor:

Comments: This version of the manuscript is much improved. However, the discussion section is long and redundant and needs to be pared down.

Answer: Without changing the original intention, the discussion section has been pared down by approximately 20%.

Replies to Reviewer 2:

None.